\begin{table}%t1
      \caption{\label{table.nonmayrit}X-ray stars not tabulated in the Mayrit catalogue 
      (Caballero~\cite{Caballero2008c})\tablefootmark{a}.}  
\par
  %%\centerline
\par
{\small
  \begin{tabular}{lcl cc cccc l}
            \hline
            \hline
            \noalign{\smallskip}
No.	 & 	 &  Name			 &  $\alpha$ 	 &  $\delta$	 &  $i$			 &  $J$ 			 &  $H$			 &  $K_{\rm s}$		 &  References\tablefootmark{c}	\\  
	 & 	 & 			 &  (J2000) 	 &  (J2000) 	 &  [mag]			 &  [mag]			 &  [mag]			 &  [mag]			 &  			\\  
            \noalign{\smallskip}
            \hline
            \noalign{\smallskip}
25	 &  *	 &  Mayrit 3020 AB	 &  05 38 44.84 	 &  --02 35 57.1 	 &  ....			 &    $10.4\pm0.2$   &   $10.70\pm0.07$ 	 &  $10.480\pm0.010	$ &  vLO03, Ca05, Bo09	\\ % Class~II		      \\  
31	 &  *	 &  [W96] 4771--1056	 &  05 39 00.52 	 &  --02 39 39.0 	 &  $12.371\pm0.06	$ &  $11.665\pm0.028$   &  $11.221\pm0.024$   &  $11.110\pm0.022$   &  Wo96, Sk08		\\ % No seq., Li~{\sc i}?     \\  
39	 &  *	 &  Mayrit 168291 AB	 &  05 38 34.31 	 &  --02 35 00.0 	 &  $12.272\pm0.03	$ &  $11.216\pm0.031$   &  $10.565\pm0.033$   &  $10.354\pm0.030$   &  He07, Ga08, Sk08  \\ % Li~{\sc i}	      \\  
46	 & 	 &  Mayrit 1093033	 &  05 39 24.56 	 &  --02 20 44.1 	 &  $12.727\pm0.16	$ &  $11.371\pm0.026$   &  $10.778\pm0.024$   &  $10.554\pm0.023$   &  He07			\\ % Candidate, no new	      \\  
47	 &  *	 &  Mayrit 68229		 &  05 38 41.35 	 &  --02 36 44.4 	 &  $14.404\pm0.03 $   &  $12.988\pm0.026$   &  $12.330\pm0.024$   &  $12.084\pm0.025$   &  Wo96, Ca07b, Sk08 \\ % Candidate	      \\    
48	 & 	 &  Mayrit 172264		 &  05 38 33.35 	 &  --02 36 17.6 	 &  $13.393\pm0.02 $   &  $12.052\pm0.027$   &  $12.295\pm0.023$   &  $11.107\pm0.027$   &  Sh04, Sk08		\\ % Candidate	      \\  
51	 & 	 &  [SWW2004] 166 	 &  05 38 53.06 	 &  --02 38 53.6 	 &  $12.717\pm0.02	$ &  $11.625\pm0.026$   &  $11.034\pm0.026$   &  $10.828\pm0.025$   &  Sh04, Ol06, Sk08  \\ % NM		      \\  
57	 &  *	 &  Mayrit 492211 	 &  05 38 27.74 	 &  --02 43 00.9 	 &  $13.636\pm0.03	$ &  $11.189\pm0.030$   &  $11.447\pm0.024$   &  $10.287\pm0.024$   &  Sh04, Sa08		\\ % Li~{\sc i}	      \\  
58	 &  *	 &  Mayrit 21023 		 &  05 38 45.31	 &  --02 35 41.3 	 &  ....			 &   $13.41\pm0.09	$ &   $12.98\pm0.06	$ &   $12.73\pm0.09	$ &  Ca07b, Bo09		\\ % Candidate  	      \\  
85	 & 	 &  Mayrit 270196 	 &  05 38 39.72 	 &  --02 40 19.7 	 &  $15.489\pm0.05	$ &  $13.746\pm0.031	$ &  $13.099\pm0.026	$ &  $12.883\pm0.028	$ &  Sk08			\\ % Candidate  	      \\  
95$^b$	 & 	 &  Mayrit 605079 	 &  05 39 24.35 	 &  --02 34 01.3 	 &  $14.501\pm0.19	$ &  $12.978\pm0.030	$ &  $12.272\pm0.026	$ &  $12.058\pm0.021	$ &  Sh04, Sa07, Sa08	\\ % Possible		      \\  
98	 & 	 &  Mayrit 1178039 	 &  05 39 33.78 	 &  --02 20 39.8 	 &  $13.783\pm0.15	$ &  $12.367\pm0.026	$ &  $11.598\pm0.023	$ &  $11.429\pm0.023	$ &  Sh04, Ol06	 	\\ % Candidate  	      \\  
99	 & 	 &  Mayrit 957055 	 &  05 39 37.29 	 &  --02 26 56.7 	 &  $13.000\pm0.03	$ &  $11.698\pm0.026	$ &  $10.974\pm0.024	$ &  $10.773\pm0.021	$ &  Sh04			\\ % Candidate  	      \\  
            \hline
         \end{tabular}}
\tablefoot{\tablefoottext{a}{Stars marked with an asterisk, ``*'', are commented in
Sect.~\ref{section.notes.table1}.}
\tablefoottext{b}{See Sect.~\ref{section.clustermembers} for a discussion on
Mayrit~605079.} 
\tablefoottext{c}{Reference abbreviations --
Wo96: Wolk (\cite{Wolk1996});
vLO03: van~Loon \& Oliveira (\cite{van2003});
Sh04: Sherry et~al. (\cite{Sherry2004});
Ca05: Caballero (\cite{Caballero2005});
Ol06: Oliveira et~al. (\cite{Oliveira2006});
Ca07b: Caballero (\cite{Caballero2007b});
Sa07: Sacco et~al. (\cite{Sacco2007});
He07: Hern\'andez et~al. (\cite{Hernandez2007});
Ga08: Gatti et~al. (\cite{Gatti2008});
Sa08: Sacco et~al. (\cite{Sacco2008});
Sk08: Skinner et~al. (\cite{Skinner2008});
Bo09: Bouy et~al. (\cite{Bouy2009}).}}\vspace*{3mm}
   \end{table}